Author’s response to reviews

Title: Miliary brain metastasis presenting with calcification in patient with lung cancer: a case report

Authors:

Minehiko Inomata (cxhcw543@yahoo.co.jp)
Ryuji Hayashi (htaro@med.u-toyama.ac.jp)
Kenta Kambara (k_kent@cotton.ocn.ne.jp)
Seisuke Okazawa (okazawa-sin@umin.net)
Shingo Imanishi (finecrackles@hotmail.com)
Tomomi Ichikawa (tomichz@gmail.com)
Kensuke Suzuki (khiro@tam.ne.jp)
Toru Yamada (tyamada-tym@umin.ac.jp)
Toshiro Miwa (mtoshi-tym@umin.ac.jp)
Tatsuhiko Kashii (tatsuk-tym@umin.ac.jp)
Shoko Matsui (shoko-tym@umin.ac.jp)
Masakiyo Sasahara (sasahara@med.u-toyama.ac.jp)
Kazuyuki Tobe (tobe@med.u-toyama.ac.jp)

Version: 3 Date: 13 June 2012

Author’s response to reviews: see over
13/June/2012

To The Journal of Medical Case Reports Editorial Team,

I am glad to hear from you and read the reviewer’s and the associate editor’s comments. I revised our manuscript according to their comments as below. I also asked the professional copyediting service to correct our manuscript. It is our pleasure for you to read our revised manuscript and consider its publication on The Journal of Medical Case Reports. Thank you and hope a good reply.

Correction:

To reviewer 1:

Thank you for your comments and appreciation. We added the some speculations on the mechanisms of tumor calcification at discussion session, last paragraph (page 9, line 2-10). In addition to that, we added two reference reports explaining these mechanisms at reference session (page 14, ref. No. 16 and 17).

To the associate editor:

First, we added the speculations as above. Second, the some grammatical errors of our first manuscript were corrected by professional copyediting service and we attached the certificate, too. These corrections are not highlighted.

To the editorial team:

1. We changed the title for matching with the journal format (title page).
2. We deleted the dates in the first manuscript. In abstract session, page 2, line 5-6. In case report session, page 4 line 11 and page 5 line 1.
3. We asked the professional copyediting service to correct our manuscript and attached the certificate. These corrections are not highlighted for avoiding confusion.

Sincerely,

Dr. Ryuji Hayashi M.D., Ph.D.

First Department of Internal Medicine,
University of Toyama,
2630 Sugitani,
Toyama City,
930-0194, Japan
Email: htar@med.u-toyama.ac.jp